Author’s response to reviews

Title: Prevalence and Causes of Blindness, Visual Impairment Among Different Ethnical Minority Groups In Xinjiang Uygur Autonomous Region, China

Authors:

Yanping Li (5861623@qq.com)
Wenyong Huang (1220792380@qq.com)
Qoyun Qiqige (aoyunqiqg@163.com)
Hongwei Zhang (zhwzhw.123@163.com)
Ling Jin (lingjin41@yahoo.com)
Pula Ti (anrabc@126.com)
Jennifer Yip (jennifer.yip@lshtm.ac.uk)
Baixiang Xiao (xiaobaixiang2006@126.com)

Version: 5 Date: 01 Feb 2018

Author’s response to reviews:

thanks for your letter dated on 31st Jan to request minor revision of the manuscript. Belowing changes has been made for your consideration now of acceptance.

1) The website linkage to the DATA.

2) The statement of using the map of China with mark of the two study sites.

Best

Xiao